Author's response to reviews

Title: Oseltamivir for treatment and prevention of pandemic influenza A/H1N1 virus infection in households, Milwaukee, 2009

Authors:

Edward Goldstein (egoldste@hsph.harvard.edu)
Benjamin C Cowling (bcowling@hku.hk)
Justin J O'Hagan (justinohagan@gmail.com)
Leon Danon (leon.danon@gmail.com)
Victoria Fang (vickyst.fang@gmail.com)
Angela Hagy (amhagy@milwaukee.gov)
Joel C Miller (joel.c.miller@gmail.com)
David Reshef (dnreshef@mit.edu)
James Robins (robins@hsph.harvard.edu)
Paul Beidrzycki (pbiedr@milwaukee.gov)
Marc Lipsitch (mlipsitc@hsph.harvard.edu)

Version: 3 Date: 23 June 2010

Author's response to reviews: see over
Dear Editors,

We’ve corrected a sentence in the Discussion according to the referee’s suggestion, changing it from “reduce, if not fully eliminate, the extent of” to “further reduce”.

Our study, which used anonymized data on confirmed influenza cases and their household contacts, was reviewed by the Human Subjects Committee at the Harvard School of Public Health and was designated as “not human subjects” (protocol # 17770-102).